Author's response to reviews

Title: case report: Situs inversus totalis with perforated duodenal ulcer. Mohammad Tayeb* MBBS, FCPS, Faiz Mohammad Khan* MBBS, FRCS, Fozia Rauf** MBBS, FCPS

Authors:

mohammad tayeb (drmtayeb@yahoo.com)
faiz mohammad khan (faizkhan.surg@yahoo.com)
fozia rauf (farauf@yahoo.com)

Version: 2 Date: 30 November 2010

Author's response to reviews:

Cover letter
To
The JMCR Editorial Team
Subject: Making the suggested changes by reviewers in our Article
Case report: Situs inversus totalis with perforated duodenal ulcer. Mohammad Tayeb* MBBS, FCPS, Faiz Mohammad Khan* MBBS, FRCS, Fozia Rauf** MBBS, FCPS
MS: 1705422835458055

1: Zakaria Habib suggested a change in the conclusion of the article which is quite right and I made the suggested changes in revised manuscript.
2: Sudha Thiruvengadam suggested that abbreviation like BP should be expanded for the first time and I did that in the revised manuscript.

I am grateful to the reviewers and editorial team for their quick response, valuable suggestions and help regarding publication of this article.

With Best Wishes
Mohammad Tayeb